Author's response to reviews

Title: A Pilot Study: The Development of a Culturally Tailored Malaysian Diabetes Education Module (MY-DEMO) based on the Health Belief Model.

Authors:

Badariah Ahmad (badariah.ahmad@monash.edu)
Amutha Ramadas (amutha.ramadas@monash.edu)
Quek KF (quek.kia.fatt@monash.edu)
Anuar Zaini Md Zain (anuar.zaini@monash.edu)

Version: 4  Date: 28 February 2014

Author's response to reviews:

Dr Badariah Ahmad,
Senior Lecturer,
Jeffrey Cheah School Of Medicine,
Monash University Sunway campus,
28th February 2014.

Dear Editor,

MS: 7617367971011927
The Development and Validation of a Culturally Tailored Malaysian Diabetes Education Module (MY-DEMO) based on the Health Belief Model (HBM).
Badariah Ahmad, Amutha Ramadas, Quek KF and Anuar Zaini Md Zain

Please find the revised article for resubmission. Please note that due to the consideration from the reviewer, the authors have made some modifications from the original title above to the following:- “A Pilot Study: The Development of a Culturally Tailored Malaysian Diabetes Education Module (MY-DEMO) based on the Health Belief Model”.

The authors are grateful for the constructive criticisms given by both reviewers and have taken the necessary steps to revised the article to the best of our ability. In addition, the authors have also included a detailed tabled response to the queries and issues which the reviewers had identified.

All the authors (B Ahmad, A Ramadas, Quek KF and A Zaini Md Zain) have made a significant contribution to the final manuscript

We look forward to your positive response and a successful publication in the near future.

Yours sincerely,
(Badariah Ahmad)